Author's response to reviews

Title: The impact of health and Islamic school-based teaching programs on smoking: a randomized control trial with 11 to 14 year olds in Aceh, Indonesia

Authors:

Teuku Tahlil (tahl0001@flinders.edu.au)
Paul R Ward (paul.ward@flinders.edu.au)
Richard J Woodman (richard.woodman@flinders.edu.au)
John Coveney (headfpppc@flinders.edu.au)

Version: 2 Date: 12 October 2012

Author's response to reviews:

Dear Editor,

As follow-up to our previous emails, I am herewith resending you the revised version of our manuscript, "The impact of health and Islamic school-based teaching programs on smoking: a randomized control trial with 11 to 14 year olds in Aceh, Indonesia".

In this version, we have added the registration number of the study as requested.

Thank you in advance. We look forward to hearing from you.

Sincerely yours

Teuku Tahlil